Author’s response to reviews

Title: Colonization of pediatric cancer patients with Candida species does not lead to elevated levels of (1->3)-beta-D-glucan, Candida mannan or Candida DNA in serum samples

Authors:

Eiman Mokaddas (e.mokaddas@hsc.edu.kw)
Mona H. Burhamah (bufahad86@hotmail.com)
Zia U. Khan (zkhan@hsc.edu.kw)
Suhail Ahmad (suhail_ah@hsc.edu.kw)

Version: 5 Date: 2 September 2010

Author’s response to reviews:

Dear Editor,

Some minor editorial corrections have been made.